Author's response to reviews

Title: The factorial structure of the mini mental state examination (MMSE) in Japanese dementia patients

Authors:

  kenta shigemori (kenta-s@seirei.ac.jp)
  Shohei Ohgi (shohei-o@seirei.ac.jp)
  Eriko Okuyama (eriko0722@tea.ocn.ne.jp)
  Takaki Shimura (tshimura@jsed.sakura.ne.jp)
  Eric Schneider (eschneid@jhsph.edu)

Version: 3 Date: 22 May 2010

Author's response to reviews: see over
May 22, 2010

The BioMed Central Editorial Team

Dear Natalie

MS: 1276675642347073; The factorial structure of the mini mental state examination (MMSE) in Japanese dementia patients

Thank you for your kind e-mail of May 10, 2010 and for the reviewers’ comments concerning our manuscript. We have revised the manuscript according to the reviewers’ suggestions. Our responses to the reviewers’ comments and the modifications we have made attached. We hope that these revisions are satisfactory and the Revised Version will be acceptable in BMC Geriatrics.

Yours Sincerely,

Kenta shigemori, RPT, MS
Graduate School of Health Sciences, Seirei Christopher University
3453Mikatahara-Cho, Hamamatsu-City Shizuoka, 433-8558, Japan
Phone, +81-53-439-1400
Fax, +81-53-439-1406
Mail address: kenta-s@seirei.ac.jp